Author's response to reviews

Title: Protocol for a pragmatic randomised controlled trial to evaluate effects of a brief intervention for emergency department attendees who present with moderate and high levels of non-specific psychological distress: a pilot study.

Authors:

Petra Lawrence (petra_lawrence@health.qld.gov.au)
Paul Fulbrook (paul.fulbrook@acu.edu.au)

Version: 11 Date: 8 July 2015

Author's response to reviews:

Dear Editor.

1. We had to change this study to a pilot study to test the efficacy of a BI. This is to inform whether a larger study may be feasible in the future. So, the endpoint is changed to whether the participant in the intervention group sought any assistance for their stress levels.

thank you

Petra Lawrence